Author's response to reviews

Title: Large cell non-Hodgkin's lymphoma masquerading as a renal carcinoma with IVC thrombosis: A case report

Authors:

Erika E. Samlowski (esamlows@artsci.wustl.edu)
Christopher Dechet (christopher.dechet@hci.utah.edu)
Alan Weissman (aweissman@nvcancer.org)
Wolfram Samlowski (wsamlowski@nvcancer.org)

Version: 4 Date: 12 February 2011

Author's response to reviews: see over
Thank you for allowing us to revise our manuscript now entitled “Large cell non-Hodgkin lymphoma masquerading as a renal carcinoma with IVC thrombosis: A case report”. We appreciate the editor’s comments and we have addressed them in the revised manuscript attached. A point-by-point reply is provided below.

1) Reference to “angiotropic large cell lymphoma” has been changed to the more contemporary “intravascular large B-cell lymphoma” in the text.
2) The apostrophe has been taken off “Hodgkin's”. It is “non-Hodgkin lymphoma.”
3) We have made very attempt to insure that formatting matches the template provided for the journal.

We hope these revisions will allow you to consider publication of our case report.

Sincerely,

Wolfram Samlowski, MD
Professor and Head
Section of Melanoma, Renal Cancer and Immunotherapy